%% Copernicus Publications Manuscript Preparation Template for LaTeX Submissions
%% ---------------------------------
%% This template should be used for copernicus.cls
%% The class file and some style files are bundled in the Copernicus Latex Package which can be downloaded from the different journal webpages.
%% For further assistance please contact the Copernicus Publications at: publications@copernicus.org
%% http://publications.copernicus.org


%% Please use the following documentclass and Journal Abbreviations for Discussion Papers and Final Revised Papers.


%% 2-Column Papers and Discussion Papers
\documentclass[gmd, manuscript]{copernicus}



%% Journal Abbreviations (Please use the same for Discussion Papers and Final Revised Papers)

% Atmospheric Chemistry and Physics (acp)
% Advances in Geosciences (adgeo)
% Advances in Statistical Climatology, Meteorology and Oceanography (ascmo)
% Annales Geophysicae (angeo)
% ASTRA Proceedings (ap)
% Atmospheric Measurement Techniques (amt)
% Advances in Radio Science (ars)
% Advances in Science and Research (asr)
% Biogeosciences (bg)
% Climate of the Past (cp)
% Drinking Water Engineering and Science (dwes)
% Earth System Dynamics (esd)
% Earth Surface Dynamics (esurf)
% Earth System Science Data (essd)
% Fossil Record (fr)
% Geographica Helvetica (gh)
% Geoscientific Instrumentation, Methods and Data Systems (gi)
% Geoscientific Model Development (gmd)
% Geothermal Energy Science (gtes)
% Hydrology and Earth System Sciences (hess)
% History of Geo- and Space Sciences (hgss)
% Journal of Sensors and Sensor Systems (jsss)
% Mechanical Sciences (ms)
% Natural Hazards and Earth System Sciences (nhess)
% Nonlinear Processes in Geophysics (npg)
% Ocean Science (os)
% Primate Biology (pb)
% Scientific Drilling (sd)
% SOIL (soil)
% Solid Earth (se)
% The Cryosphere (tc)
% Web Ecology (we)



%% \usepackage commands included in the copernicus.cls:
%\usepackage[german, english]{babel}
%\usepackage{tabularx}
%\usepackage{cancel}
%\usepackage{multirow}
%\usepackage{supertabular}
%\usepackage{algorithmic}
%\usepackage{algorithm}
%\usepackage{float}
%\usepackage{subfig}
%\usepackage{rotating}


\begin{document}
\begin{thebibliography}{}

\bibitem[{{ALADIN International Team}(1997)}]{aladin}
ALADIN International Team:, The ALADIN project: Mesoscale modelling seen as a basic tool for weather forecasting and atmospheric research, WMO Bull., 46, 317--324, 1997.
\bibitem[{{Alpert et al.}(1990)}]{alpert1990}
Alpert, P., Neeman, B.U., and Y. Shay-El, Y.; Intermonthly variability of cyclone tracks in the Mediterranean, J. Climate, 3, 1474--1478, 1990.
%\bibitem[{{Baji\'{c} et al.}(2007)}]{bajic2007}
%Baji\'{c}, A., S. Ivatek-\v{S}ahdan, and K. Horvath (2007), Spatial distribution of wind speed in Croatia obtained using the ALADIN model, {\it Cro. Met. J.,} {42}, 67--77.
\bibitem[{{Boyd}(2005)}]{boyd2005}
Boyd, J.P.: Limited-Area Fourier Spectral Models and Data Analysis Schemes: Windows, Fourier Extension, Davies Relaxation, and All That. Mon. Wea. Rev., 133, 2030--2042, 2005.
\bibitem[{{Brankovi\'{c} et al.}(2007)}]{brankovic2007}
Brankovi\'{c}, \v{C}., Matja\v{c}i\'{c}, B., Ivatek-\v{S}ahdan, S., and Buizza, R.: Dynamical downscaling of ECMWF EPS forecasts applied to cases of severe weather in Croatia, ECMWF RD Technical Memorandum, No. 507, 38 pp., 2007.
\bibitem[{{Brankovi\'{c} et al.}(2008)}]{brankovic2008}
Brankovi\'{c}, \v{C}., Matja\v{c}i\'{c}, B., Ivatek-\v{S}ahdan, S., and Buizza, R.: Downscaling of ECMWF Ensemble Forecasts for Cases of Severe Weather: Ensemble Statistics and Cluster Analysis, Mon. Wea. Rev., 136, 3323--3342, 2008.
\bibitem[{{Bro\v{z}kova et al.}(2001)}]{brozkova}
Bro\v{z}kova, R., Klari\'{c}, D., Ivatek-\v{S}ahdan, S., Geleyn, J.-F., Casse, V., \v{S}irok\'{a}, M., R\'{a}dnoti, G., Janou\v{s}ek, M., Stadlbacher, K., and Seidl, H.: DFI Blending, an alternative tool for preparation of the initial conditions for LAM, PWRP Report Series No. 31, WMO-TD, No. 1064, 2001.
%\bibitem[{{Brzovi\'{c}}(1999)}]{brzovic1999}
%Brzovi\'{c}, N. (1999), Factors affecting the Adriatic cyclone and associated windstorms, {\it Contr. Atmos. Phys.,} {72}, 51--65.
%\bibitem[{{Brzovi\'{c} and Strelec Mahovi\'{c}}(1999)}]{brzomah1999}
%Brzovi\'{c}, N., and N. Strelec Mahovi\'{c} (1999), Cyclonic activity and severe jugo in the Adriatic, {\it Phys. Chem. Earth,} {24}, 653--657.
\bibitem[{{Campinis at al.}(2000)}]{campinis2000}
Campinis, J., Genoves, A., Jansa, A., Guijarro, J.A., and Ramis, C.: A catalogue and a classification of surface cyclones for the Western Mediterranean, Int. J. Climatol., 20, 969--984, 2000.
\bibitem[{{Cassou and Terray}(2001)}]{cassou2001}
Cassou, C., and Terray, L.: Oceanic forcing of the wintertime low-frequency atmospheric variability in the north atlantic european sector: a study with the arpege model, J. Climate, 14, 4266--4291, doi: http://dx.doi.org/10.1175/1520-0442(2001)014, 2001.
\bibitem[{{Courtier and Geleyn}(1988)}]{courtier1988}
Courtier, P., and Geleyn, J.-F.: A global numerical weather prediction model with variable resolution: application to the shallow water equations. Quart. J. Roy. Meteor. Soc., 114, 1321--1246, 1988.
\bibitem[{{Davies}(1976)}]{davies1976}
Davies, H.C.: A lateral boundary formulation for multi-level
prediction models. Q. J. R. Meteorol. Soc., 102, 405--418, 1976.
\bibitem[{{Davies}(1983)}]{davies1983}
Davies, H.C.: Limitations of some common lateral boundary schemes
used in regional NWP models. Mon. Wea. Rev., 111, 1002--1012, 1983.
\bibitem[{{Davies}(2014)}]{davies2014}
Davies, T.: Lateral boundary conditions for limited area models. {\it Q. J. R. Meteorol. Soc.}, 140: 185--196, 2014.
\bibitem[{{Dee et al.}(2011)}]{dee2011}
Dee, D. P., Uppala, S.M., Simmons, A.J., Berrisford, P., Poli, P., Kobayashi, S., Andrae, U., Balmaseda, M.A., Balsamo, G., Bauer, P., Bechtold, P., Beljaars, A.C.M., van de Berg, L., Bidlot, J., Bormann, N., Delsol, C.,  Dragani, R., Fuentes, M., Geer, A.J., Haimberger, L., Healy, S.B., Hersbach, H., H\'{o}lm, E.V., Isaksen, L., K\`{a}llberg, P., K\"{o}hler, M., Matricardi, M., McNally, A.P., Monge-Sanz, B.M., Morcrette, J.-J., Park, B.-K., Peubey, C., de Rosnay, P., Tavolato, C., Th\'{e}paut, J.-N., and Vitart, F.: The ERA-Interim reanalysis: configuration and performance of the data assimilation system. {\it Q. J. R. Meteorol. Soc.}, {137}, 553--597. doi: 10.1002/qj.828, 2011.
\bibitem[{{Degrauwe et al.}(2012)}]{deg2012}
Degrauwe, D., Caluwaerts, S., Voitus, F., Hamdi, R., and Termonia, P.: Application of Boyds Periodization and Relaxation Method in a Spectral Atmospheric Limited Area Model. Part II: Accuracy Analysis and Detailed Study of the Operational Impact. Mon. Wea. Rev., 140, 3149--3162, 2012. 
\bibitem[{{Denis et al.}(2002)}]{denis2002}
Denis, B., Laprise, R., Caya, D. and C\v{o}t\'{e}, J.: Downscaling ability of one-way nested regional climate models: the big-brother experiment. Climate Dynamics, 18, 627--646, 2002. 
\bibitem[{{Denis et al.}(2003)}]{denis2003}
Denis, B., Laprise, R. and Caya, D.: Sensitivity of a regional climate model to the resolution of the lateral boundary conditions. Climate Dynamics, Climate Dynamics, 20, 107--126, 2003.
\bibitem[{{De Troch et al.}(2013)}]{detroch2013}
De Troch, R., Hamdi, R., Van de Vyver, H., Geleyn, J.-F., and Termonia, P.: Multiscale Performance of the ALARO-0 Model for Simulating Extreme Summer Precipitation Climatology in Belgium. J. Climate, 26, 8895--8915, 2013. 
\bibitem[{{Doswell et al.}(1996)}]{doswell1996}
Doswell, C.A., Brooks, H.E., and Maddox, R.A.: Flash flood forecasting: An ingredients-based methodology. {\it Wea. Forecasting}, 11, 560--581, 1996.
%\bibitem[{{Guidard and Fischer}(2008)}]{guidard2008}
%Guidard, V., and Fischer, C.: Introducing the coupling information in a limited-area variational assimilation. {\it Q. J. R. Meteorol. Soc.}, 134, 723--735i, 2008.
\bibitem[Hamdi et al.(2012)]{hamdi2012}
Hamdi, R., Van de Vyver, H. and Termonia, P.: New cloud and microphysics parameterisation for use in high-resolution dynamical downscaling: application for summer extreme temperature over Belgium. Int. J. Climatol., 32: 2051--2065. doi: 10.1002/joc.2409, 2012.
\bibitem[Hamdi et al.(2014)]{hamdi2014}
Hamdi, R., Van de Vyver, H., De Troch, R. and Termonia, P.: Assessment of three dynamical urban climate downscaling methods: Brussels's future urban heat island under an A1B emission scenario. Int. J. Climatol., 34: 978--999. doi: 10.1002/joc.3734, 2014.
\bibitem[{{Haugen and Machenhauer}(1993)}]{haugen1993}
Haugen, J., and Machenhauer, B.:  A spectral limited-area
formulation with time-dependent boundary conditions applied
to the shallow-water equations. Mon. Wea. Rev., 121, 2618--2630, 1993.
%\bibitem[{{Holton}(1992)}]{holton1992}
%Holton, J.R. (1992), An Introduction to Dynamic Meteorology, Third Edition,{\it Academic Press,} 511 pp.
\bibitem[Hortal(2002)]{hortal2002}
Hortal, M.: The development and testing of a new two-time-level semi-Lagrangian scheme (SETTLS) in the ECMWF forecast model. Q.J.R. Meteorol. Soc., 128: 1671--1687, 2002. 
\bibitem[{{Horvath et al.}(2011)}]{horvath2011}
Horvath, K., Baji\'{c}, A., and Ivatek-\v{S}ahdan, S.:  Dynamical downscaling of wind speed in complex terrain prone to bora-type flows, J. Appl. Meteor. Climatol., 50, 1676--1691, 2011.
\bibitem[{{Horvath et al.}(2009)}]{horvath2009}
Horvath, K., Ivatek-\v{S}ahdan, S., Ivan\v{c}an-Picek, B., and Grubi\v{s}i\'{c}, V.: Evolution and structure of two severe cyclonic bora events: contrast between the northern and southern Adriatic, Weather and forecasting, 24, 946--964, 2009.
\bibitem[{{Horvath et al.}(2008)}]{horvath2008}
Horvath, K., Lin, Y.-H., and Ivancan-Picek, B.: Classification of cyclone tracks over the Apennines and the Adriatic Sea, Mon. Wea. Rev., 136, 2210--2227, 2008.
\bibitem[{{Ivatek-\v{S}ahdan and Ivan\v{c}an-Picek}(2006)}]{ivateks2006}
Ivatek-\v{S}ahdan, S. and Ivan\v{c}an-Picek, B.: Effects of different initial and boundary conditions in ALADIN/HR simulations during MAP IOPs, Meteorol. Z., 15, 187--197, 2006.
\bibitem[{{Ivatek-\v{S}ahdan and Tudor}(2004)}]{ivateks2004}
Ivatek-\v{S}ahdan, S., and Tudor, M.: Use of high-resolution dynamical adaptation in operational suite and research impact studies, Meteorol Z, 13(2), 1--10, 2004.
\bibitem[{{Laprise}(2003)}]{laprise2003}
Laprise, R.: Resolved scales and nonlinear interactions in limited-area models, J. Atmos. Sci., 60, 768--779, 2003.
\bibitem[{{Laprise}(2008)}]{laprise2008}
Laprise, R., de El\'{i}a, R., Caya, D., Biner, S., Lucas-Picher, P., Diaconescu, E., Leduc, M., Alexandru, A., Separovic, L., Canadian Network for Regional Climate Modelling and Diagnostics: Challenging some tenets of regional climate modelling. Meteorol. Atmos. Phys., 100, 3--22, 2008.
\bibitem[{{Lionello et al.}(2006)}]{lionello2006}
Lionello, P., Bhend, J., Buzzi, A., Della-Marta, P. M., Krichak, S. O., Jans\'{a}, A., Maheras, P.,
Sanna, A., Trigo, I. F., and Trigo, R.: Cyclones in the Mediterranean region: climatology
and effects on the environment, in: Mediterranean Climate Variability, edited by: Lionello, P.,
Malanotte-Rizzoli, P., and Boscolo, R., 325--372, Elsevier, 2006. 
\bibitem[{{Lynch and Huang}(1992)}]{lynch1992}
Lynch, P., and Huang, X-Y.: Initialization of the HIRLAM model using a digital filter. Mon. Wea. Rev, 120, 1019--1034, 1992.
\bibitem[{{Lynch}(1997)}]{lynch1997a}
Lynch, P.: The Dolph--Chebyshev Window: A Simple Optimal Filter. Mon. Wea. Rev., 125, 655--660, 1997.
\bibitem[{{Lynch et al.}(1997)}]{lynch1997b}
Lynch, P., Giard, D., and Ivanovici, V.: Improving the Efficiency of a Digital Filtering Scheme for Diabatic Initialization. Mon. Wea. Rev., 125, 1976--1982, 1997.
%\bibitem[McDonald (1999)]{mcdonald1999}
%McDonald, A.:  A review of lateral boundary conditions for
%limited-area forecast models. Proc. Indian Nat. Sci. Acad., 65,
%91--105, 1999.
\bibitem[Nicolis(2007)]{nicolis2007}
Nicolis, C.: Dynamics of model error: The role of the
boundary conditions. J. Atmos. Sci., 64, 204--215, 2007.
\bibitem[Nutter et al.(2004)]{nutter2004}
Nutter, P., Stensrud, D., and Xue, M.:  Effects of coarsely resolved and temporally interpolated lateral boundary conditions on the dispersion of limited-area ensemble forecasts.
Mon. Wea. Rev., 132, 2358--2377, 2004.
\bibitem[R\'{a}dnoti (1995)]{radnoti1995}
R\'{a}dnoti, G.: Comments on  A spectral limited-area formulation with time-dependent boundary conditions applied to the shallow-water equations. Mon. Wea. Rev., 123, 3122--3123, 1995.
\bibitem[Simmons and Burridge (1981)]{simmons1981}
Simmons, A.J., and Burridge, D.M.: An Energy and Angular-Momentum
Conserving Vertical Finite-Difference Scheme and Hybrid Vertical
Coordinates. Mon. Wea. Rev. 109, 758--766, 1981.
\bibitem[{{Stane\v{s}i\'{c}}(2011)}]{stanesic2011}
Stane\v{s}i\'{c}, A.: Assimilation system at DHMZ: development and first verification results. Cro. Met. Jour., 44/45, 3--17, 2011.
\bibitem[{{Staniforth}(1997)}]{staniforth}
Staniforth, A.: Regional modelling: A theoretical discussion. Meteorol. Atmos. Phys., 63, 15-29, 1997.
\bibitem[{{Termonia}(2003)}]{termonia2003}
Termonia, P.: Monitoring and improving the temporal interpolation of lateral-boundary coupling data for limited area models. Mon. Wea. Rev., 131, 2450--2463, 2003.
\bibitem[{{Termonia}(2004)}]{termonia2004}
Termonia, P.: Monitoring of the coupling update frequency of a limited-area model by means of a recursive digital filter. Mon. Wea. Rev., 132, 2130--2141, 2004.
%\bibitem[Termonia and Hamdi (2007)]{termonia2007}
%Termonia, P., and R. Hamdi, (2007), Stability and accuracy of the physics-dynamics coupling in spectral models. {\it Q. J. R. Meteorol. Soc.}, 133, 1589--1604.
\bibitem[{{Termonia}(2008)}]{termonia2008}
Termonia, P.: Scale-selective digital filter initialization. Mon. Wea. Rev., 136, 5246--5255, 2008.
%\bibitem[Termonia and V\'{o}itus (2008)]{termonia2008b}
%Termonia, P., and F. V\'{o}itus, 2008: Externalizing the lateral boundary conditions from the dynamic core in semi-implicit semi-Lagrangian models. Tellus, 60A, 632--648.
\bibitem[Termonia et al.(2009)]{termonia2009}
Termonia, P., Deckmyn, A., and Hamdi, R.: Study of the lateral boundary condition temporal resolution problem and a proposed solution by means of boundary error restarts. Mon. Wea. Rev., 137, 3551--3566, 2009.
\bibitem[Termonia et al.(2011)]{termonia2011}
Termonia, P., Degrauwe, D., and Hamdi, R.: Improving the Temporal Resolution Problem by Localized Gridpoint Nudging in Regional Weather and Climate Models. Mon. Wea. Rev., 139, 1292--1304, 2011. 
\bibitem[Termonia et al.(2012)]{termonia2012}
Termonia, P., Voitus, F., Degrauwe, D., Caluwaerts, S., and Hamdi, R.: Application of Boyds Periodization and Relaxation Method in a Spectral Atmospheric Limited-Area Model. Part I: Implementation and Reproducibility Tests. Mon. Wea. Rev., 140, 3137--3148, 2012.
doi: http://dx.doi.org/10.1175/MWR-D-12-00033.1 
\bibitem[{{Tudor and Ivatek-\v{S}ahdan}(2010)}]{tudor2010}
Tudor, M., and Ivatek-\v{S}ahdan, S.: The case study of bura of 1$^{st}$ and 3$^{rd}$ February 2007, Meteorol. Z., 19(5), 453--466, 2010.
\bibitem[Tudor et al. (2013)]{tudor2013}
Tudor, M., Ivatek-\v{S}ahdan, S., Stane\v{s}i\'{c}, A., Horvath, K., Baji\'{c}, A.: Forecasting Weather in Croatia Using ALADIN Numerical Weather Prediction Model. In: Climate Change and Regional/Local Responses. Eds: Zhang, Y. and P. Ray, InTech, Rijeka, 59--88, 2013.
\bibitem[{{Tudor and Termonia}(2010)}]{tudor2010b}
Tudor, M. and Termonia, P.: Alternative formulations for incorporating
lateral boundary data into limited area models, Mon. Wea. Rev., 138, 2867--2882, 2010.
\bibitem[Uppala et al.(2005)]{uppala2005}
Uppala, S. M., K{\aa}llberg, P. W., Simmons, A. J., Andrae, U., Bechtold, V. D. C., Fiorino, M., Gibson, J. K., Haseler, J., Hernandez, A., Kelly, G. A., Li, X., Onogi, K., Saarinen, S., Sokka, N., Allan, R. P., Andersson, E., Arpe, K., Balmaseda, M. A., Beljaars, A. C. M., Berg, L. V. D., Bidlot, J., Bormann, N., Caires, S., Chevallier, F., Dethof, A., Dragosavac, M., Fisher, M., Fuentes, M., Hagemann, S., H\'{o}lm, E., Hoskins, B. J., Isaksen, L., Janssen, P. A. E. M., Jenne, R., Mcnally, A. P., Mahfouf, J.-F., Morcrette, J.-J., Rayner, N. A., Saunders, R. W., Simon, P., Sterl, A., Trenberth, K. E., Untch, A., Vasiljevic, D., Viterbo, P. and Woollen, J.: The ERA-40 re-analysis. Q.J.R. Meteorol. Soc., 131: 2961--3012. doi: 10.1256/qj.04.176, 2005.
\bibitem[V\'{a}nnitsem and Chome (2005)]{vannitsem}
V\'{a}nnitsem, S., and Chome, F.: One-way nested regional climate simulations and domain size.  J. Climate, 18, 229--233, 2005.
%\bibitem[von Storch et al.(2000)]{storch2000}
%von Storch, H., Langenberg, H., and Feser, F.: A spectral
%nudging technique for dynamical downscaling purposes. Mon.
%Wea. Rev., 128, 3664--3673, 2000.
%\bibitem[Waldron et al.(1996)]{waldron1996}
%Waldron, K. M., Peagle, J., and Horel, J.D.: Sensitivity of
%a spectrally filtered and nudged limited area model to outer
%model options. Mon. Wea. Rev., 124, 529--547, 1996.
\bibitem[Warner et al.(1997)]{warner1997}
Warner, T., Peterson, R., and Treadon, R.: A tutorial on lateral
boundary conditions as a basic and potentially serious limitation
to regional numerical weather prediction.  Bull. Amer. Meteor.
Soc., 78, 2599--2617, 1997.
\bibitem[Wernli et al.(2002)]{wernli2002}
Wernli, H., Dirren, S., Liniger, M.A., and Zillig, M.: Dynamical aspects of the life cycle of the winter storm Lothar (24--26 December 1999). Q. J. R. Meteorol. Soc., 128, 405--429, 2002.
\bibitem[{{\v{Z}agar et al.}(2013)}]{zagar}
\v{Z}agar, N., Honzak, L., \v{Z}abkar, R., Skok, G., Rakovec, J., and Ceglar, A.: Uncertainties in a regional climate model in the midlatitudes due to the nesting tecnique and the domain size, J. Geophys. Res. Atmos., 118, 6189--6199, 2013.

%\bibitem[AUTHOR(YEAR)]{LABEL}
%REFERENCE 1

%\bibitem[AUTHOR(YEAR)]{LABEL}
%REFERENCE 2

\end{thebibliography}


\linenumbers

\title{Methods for automatized detection of rapid changes in lateral boundary condition fields for NWP limited area models}


% \Author[affil]{given_name}{surname}

\Author[1]{Martina}{Tudor}
%\Author[]{}{}
%\Author[]{}{}

\affil[1]{Meteorological and Hydrological Service, Gri\v{c} 3, Zagreb, Croatia}
%\affil[]{ADDRESS}

%% The [] brackets identify the author with the corresponding affiliation. 1, 2, 3, etc. should be inserted.



\runningtitle{Rapid changes in LBC fields}

\runningauthor{M. Tudor}

\correspondence{Martina Tudor (tudor@cirus.dhz.hr)}



\received{}
\pubdiscuss{} %% only important for two-stage journals
\revised{}
\accepted{}
\published{}

%% These dates will be inserted by Copernicus Publications during the typesetting process.


\firstpage{1}

\maketitle



\begin{abstract}
Three hourly temporal resolution of lateral boundary data for limited area models (LAMs) can be too infrequent to resolve rapidly moving storms.
This problem is expected to be worse with increasing horizontal resolution.
In order to detect intensive disturbances in surface pressure moving rapidly through the model domain, a filtered surface pressure field (MCUF) is computed operationally in the ARPEGE global model of M\'{e}t\'{e}o France.
The field is distributed in the coupling files along with conventional meteorological fields used for lateral boundary conditions (LBCs) for the operational forecast using limited area model ALADIN (Aire Limit\'{e}e Adaptation dynamique D\'{e}veloppement InterNational) in the Meteorological and hydrological service of Croatia (DHMZ).
Here an analysis is performed of the MCUF field for the LACE coupling domain for the period since 23rd January 2006, when it became available, until 15th November 2014. The MCUF field is a good indicator of rapidly moving pressure disturbances (RMPDs). Its spatial and temporal distribution can be associated to the usual cyclone tracks and areas known to be supporting cyclogenesis.
Alternative set of coupling files from IFS operational run in ECMWF is also available operationally in DHMZ with 3 hourly temporal resolution but the MCUF field is not available. 
Here, several methods are tested that detect RMPDs in surface pressure a posteriori from the IFS model fields provided in the coupling files. 
MCUF is computed by running ALADIN on the coupling files from IFS. 
The error function is computed using one time step integration of ALADIN on the coupling files without initialization, initialized with digital filter initialization (DFI) or scale selective DFI (SSDFI).
Finally, the amplitude of changes in the mean sea level pressure is computed from the fields in the coupling files.
The results are compared to the MCUF field of ARPEGE and the results of same methods applied to the coupling files from ARPEGE.
Most methods give a signal for the RMPDs, but DFI reduces the storms too much to be detected. 
The error function without filtering and amplitude have more noise, but the signal of a RMPD is also stronger. 
The methods are tested for NWP LAM ALADIN, but could be applied to other LAMs and benefit the performance of climate LAMs.
\end{abstract}



\introduction  %% \introduction[modified heading if necessary]
Operational lateral boundary conditions (LBCs) are provided to limited area models (LAMs) at a time interval of several hours, refered to as the coupling update period. These data are used at lateral boundaries of the LAM domain every LAM time-step of several minutes. Consequently, LBC data of the large scale model are (linearly) interpolated in time. The interpolation procedure distorts the model fields and can lead to LAM forecast failures in case of fast propagating storms.
The problem of linear interpolation of model fields in time for cases with rapidly moving storms that enter the LAM domain is expected to become worse as both global models and LAMs move to higher resolutions.
These storms are associated to rapidly moving pressure disturbances that will be refered as RMPDs in this text.
The problem could be even more pronounced in climate LAM's that couple to large scale data that are available with a longer interval.

One needs LBC data to represent scales that are too large to be periodic on LAM domain \citep{laprise2003}.
Various schemes for treating LBC data suffer from different problems \citep{davies1983}.
Model errors propagate from the lateral boundaries through the domain during the forecast time \citep{nicolis2007},
these errors amplify and spread further with longer time of integration \citep{nutter2004}.
A large LAM domain was recommended \citep{staniforth} to prevent boundary induced errors from propagating to the area of interest.
However, there are problems that can not be cured by making LAM domain larger \citep{vannitsem}.
For an overview of issues related to LBCs, see \cite{warner1997}.

Regional climate models are expected to develop small scale features due to high resolution surface forcings, nonlinearities in atmospheric dynamics and hydrodynamic instabilities \citep{denis2002}.
Large coupling update interval can make LBCs act as a filter of small scale features that (should) enter the LAM domain.
Climate LAM without small scale information in the initial conditions and LBCs develop small scale variance even in the absence of surface forcing due to nonlinear cascade of variance \citep{laprise2008}, but it takes several days.

Currently, there are two sets of LBC data that can be used for operational forecast using ALADIN \citep{aladin} (Aire Limit\'{e}e Adaptation dynamique D\'{e}veloppement InterNational) LAM in Meteorological and Hydrological Service of Croatia (DHMZ).
One is from global Integrated Forecast System (IFS) of the European Centre for Medium-Range Weather Forecasts (ECMWF) and another is from the global model Action de Recherche Petite Echelle Grande Echelle (ARPEGE, see eg. \cite{cassou2001}) of Meteo France.
The LBCs from the global numerical weather prediction (NWP) models ARPEGE and IFS are operationally provided with a 3 hour interval. 
These are used for running the operational ALADIN forecast in 8km resolution \citep{tudor2013}. 
Coupling is performed along the lateral boundaries in the 8 gridpoints from domain edge by means of \cite{davies1976} coupling scheme and using linear interpolation in time of the input fields from the global model. 

\citet{termonia2003} has analysed the Lothar storm \citep{wernli2002} and found that
the three hourly coupling update interval is insufficient for resolving the storm in lateral boundaries.
Also, \cite{davies2014} finds that 3 hourly LBCs lose information for 12km resolution LAM coupled to 12km resolution large scale model (see Figure 5c in \cite{davies2014}).
In order to monitor the occurrence of potential LAM forecast failures due to insufficient coupling update frequency, a recursive high-pass filter \citep{termonia2004} has been implemented to the ARPEGE model and applied to the surface pressure field. 
The filtered surface pressure field is refered to as monitoring of the coupling update frequency (MCUF) field.
Large values of the MCUF field indicate a RMPD in the surface pressure through that model grid point. 
A value larger than a threshold value suggests that a fast cyclone has moved through the area.

The MCUF field is provided  since 06 UTC run on 23rd January 2006 in the coupling files from global model ARPEGE, run operationally in Meteo France, for the common coupling domain used for LBC data in 6 countries (Austria, Croatia, Czech Republic, Hungary, Slovakia and Slovenia). 
This common domain will be refered to as the LACE domain (Limited Area for Central Europe).
The horizontal resolution of the LACE coupling domain provided from ARPEGE has changed over the years (see Table \ref{table}) but the aerial coverage of the LACE coupling domain provided from ARPEGE remained te same (see the aerial coverage of the green isolines in Figure \ref{mslp1}). 
Local operational domains are smaller than the LACE domain, but have higher horizontal resolution and have coupling zones 8 gridpoints wide along lateral boundaries.
If the point with the large MCUF value is inside the coupling zone of the ALADIN domain,
it can be expected that the ALADIN model run will miss the cyclone strength
due interpolation of boundary data in time.
These events are expected to be rare, at least according to the analysis performed on one year of data for the Belgian domain \citep{termonia2009}. 
But rapid changes in surface pressure are associated to the most intensive storms moving rapidly, pose a threat to the public and require warning. 
It is very important that operational NWP models forecast such events.
The frequency of such events is analysed for the LACE domain on almost 9 years of data from the operational ARPEGE fields (since 23rd January 2006 until 15th November 2014).

The most obvious solution to this problem is to increase the frequency of the available LBC data and most of the centres that run both global models and LAMs use hourly input fields for the LAMs. However, this solution is not very practical for the meteorological services that run only LAMs and rely on LBC data from somewhere else.
On the other hand, if 3 hourly data is insufficient for global model run in roughly 16 km resolution and LAM in 8 km resolution, then hourly data would be less satisfactory when both global model and LAM move to higher resolutions (as was already announced at various meetings in 2014). Also, running old cases from stored archive data requires using LBCs with 3 hours interval.

There are other solutions proposed to solve the problem of errors in LBCs caused by time interpolation of fields. 
The first one \citep{termonia2009} is to restart
the model forecast from the coupling file when the storm is inside the domain using
the scale selective digital filter initialization \citep{termonia2008}.
The second one is to insert the storm by means of gridpoint nudging \citep{termonia2011}. 
Both of these require to stop the model run, insert the storm artifically and continue the model run from there.
Using corrected interpolation with time derivatives \citep{termonia2003}, Boyd's periodization method \citep{boyd2005,termonia2012} can also improve the forecast \citep{deg2012}, and alternative methods of interpolating LBC data in time \citep{tudor2010b} do not require restarts, but are computationally expensive, so these would also be used only when needed.
However, in order to apply any of these solutions, we should first detect the RMPD in the fields used on lateral boundaries.

Using MCUF implies that the global model computes it operationally and distributes the field in the output files together with the other forecast fields.
However, LAM can be coupled to various global model forecasts or larger scale LAMs for operational forecast, and re-analyses for climate model studies or simulations of specific phenomena. 
With the exception of ARPEGE, global models do not provide a field that would diagnose rapid changes in pressure that occured in each grid-point during a time interval between two consecutive output files.
The centers that provide global model fields could be discouraged to compute MCUF field due to computational cost and potentially complex implementation in the model code, and especially to re-run the re-analysis cycles to provide such data for studies of historical weather.
It is therefore usefull to detect RMPDs a posteriori using the standard meteorological fields usually provided in model output.
The method should enable automatic detection of a RMPD to be usefull in the operational forecast as well as in the climate simulations using LAM.
As pointed by the reviewers, fast-moving disturbances in the upper layers of the atmosphere or inertia-gravity waves are more common. These are also a source of error in LAMs while MCUF detects disturbances in the surface pressure.
The focus of this article are rapidly moving disturbances in surface pressure, but a method that detects them could be applied to an upper level field.

LAMs used for simulations of climate use input LBCs that are available in coupling update interval of 3 hours or more.
Simultaneously, LAMs tend towards higher horizontal resolutions.
A number of climate studies has been performed \citep{horvath2011,hamdi2012,detroch2013,hamdi2014} using ALADIN in combination with ERA40 \citep{uppala2005} and ERAInterim \citep{dee2011} datasets for LBCs.
These applications would also benefit from a method that would detect RMPDs a-posteriori from the standard meteorological fields used for LBC. 

The NWP suite at DHMZ is focused on forecasting weather on the area of Croatia. 
Cyclones that affect that area often originate from western Mediterranean and Adriatic \citep{horvath2008,horvath2009} that is recognized as a particularly active region with respect to cyclones \citep{campinis2000,alpert1990}.
Severe precipitation events occur when cyclone produces convergence of the moist air and a large quantity of precipitable water \citep{lionello2006}.
Western Mediterranean experiences flash flood events that arise from extremely high rainfall rates \citep{doswell1996}.

The MCUF field is not provided in the LBC files of IFS provided by ECMWF. 
On 1st January 2014 the operational ALADIN forecast in DHMZ has switched to using IFS coupling files. 
It is possible to compute MCUF field by running ALADIN on the resolution and domain of the coupling fields.
Here an analysis is performed of the MCUF field computed by running ALADIN for the common LACE coupling domain for the files provided from IFS since 27th October 2010 until 15th November 2014.
Otherwise, it is possible to estimate the error that arises due to linear interpolation of LBC data in time \citep{termonia2003} from model tendencies obtained by running ALADIN for one time step. 
The error was estimated for surface pressure and mean sea level pressure using coupling data without initialization, or initialized to remove the high frequency noise.
Additionally, this work proposes to estimate the magnitude of pressure variations by computing a simple amplitude of oscillations between the successive coupling files.

The next section describes the models briefly, the methods used to detect RMPDs and the effect of linear interpolation in time on mean sea level pressure.
The analysis of 9 years of the MCUF field from ARPEGE is presented in Section 3.
Results of methods for detecting RMPDs in IFS coupling fields are presented in Section 4.
The last section gives conclusions.

\section{Model description and metods of detection of rapidly moving pressure disturbances}

\subsection{Operational forecast model}

ALADIN is used for operational weather forecast in DHMZ in 8 km resolution using hydrostatic dynamics, 2-time-level semi-implicit semi-Lagrangian and stable extrapolation two-time-level scheme \citep{hortal2002}. Operationally, the model uses 37 levels in the vertical and a mass-based hybrid terrain-following vertical coordinate $\eta$ \citep{simmons1981}. 

The initial conditions for the operational forecast are obtained using data assimilation procedure \citep{stanesic2011}.
Details of the operational forecast suite as well as model set-up are provided in \citet{tudor2013}, but there were few changes.
The forecast is run up to 72 hours four times a day, starting from 00, 06, 12 and 18 UTC analyses,
 and coupled to LBC fields from IFS in delayed mode. This means that LBC for 6 hour forecast from 18 UTC run of IFS is used for initial LBC for 00 run of the next day, 9 hour forecast from 18 UTC run of IFS is used for 3 hour forecast LBC for 00 run of the next day, and so on.

The 8km resolution operational forecast is coupled to a global model on the 8 points wide zone alnog lateral boundaries using relaxation tecnique \citep{davies1976} and linear interpolation of LBC data in time \citep{haugen1993,radnoti1995}.
Each coupling file contains the complete set of fields needed to initialize the ALADIN model forecast.

Digital filter initialization (DFI) is implemented in ALADIN in order to remove high-frequency noise \citep{lynch1992} that arise due to interpolation of the coupling fields from the global model grid to the grid of the coupling files and then again to the resolution of the LAM (and changes in height of topography in different models/resolutions).
Since DFI can considerably reduce the depth of the RMPD due to the Doppler effect, alternative scale selective digital filter initialization (SSDFI) was proposed, implemented and tested in the ALADIN model \citep{termonia2008}.

\subsection{Global model ARPEGE}

ARPEGE is a global sem-lagrangian spectral model run operationally at Meteo France on a stretched and rotated grid \citep{courtier1988} with highest horizontal resolution over France and lowest resolution on the opposite side of the Earth. 
The horizontal resolutions in the model forecast and data assimilation procedure were changing during the 9 years when the MCUF field was computed in the operational ARPEGE forecast. 
The horizontal resolution of the coupling files also changed twice, see Table \ref{table}.

ARPEGE can use coarser resolution in variational data assimilation procedure than in the forecast run.
The fields from the operational forecast are interpolated from the stretched and rotated native model grid to the grid of the limited area LACE domain in Lambert projection of the coupling files.

The fields from operational ARPEGE forecasts are available in the coupling files with 3 hour interval for 4 runs per day (starting from 00, 06, 12 and 18 UTC analyses) and extending up to 72 for the 00, 06 and 12 UTC runs and up to 60 hours for the 18 UTC run.
ARPEGE computes the MCUF field operationally according to \cite{termonia2004} and the field is distributed in the coupling files.

\subsection{Global model IFS}

IFS is also a global spectral model that uses semi-Lagrangian advection. It is run oparationally at ECMWF with uniform horizontal resolution over the globe.
The details of the operational set-up in the model forecast and data assimilation have changed over the years used for this study, while the LBC files were available operationally, as did the operational model versions.
The model forecast fields are interpolated from the IFS model grid to the LAM grid in Lambert projection and the horizontal resolution of the coupling files remained 15.4 km (see Table \ref{table}).

Following the research studies where LBC data from IFS has been used for studies of severe weather cases \citep{ivateks2006,brankovic2007,brankovic2008}, the operational forecast run of the ALADIN model in DHMZ has switched to uding LBC data from IFS on 1st January 2014.

The MCUF field is not computed by the IFS operational suite and therefore not available in the coupling files from IFS provided by ECMWF. 
Rapid changes in the surface pressure or the mean sea level pressure were detected in the fields provided from IFS operational forecast in the coupling files on the LACE common domain using a number of tools.

\begin{itemize}
\item ALADIN was run on the LACE domain (in the resolution of the coupling files) with 600 seconds time step and the MCUF field was computed during the model run. The computed MCUF field will be referred to as IFSM. However, this means that a different model was run (different dynamics and physics) and the results can be different than when computed in the host model. 
\item The error function from \cite{termonia2003} was computed by running one time-step forecast starting from fields in the coupling files (in the same horizontal and vertical resolution), three sets of experiments were performed using initialization without filtering, using DFI or SSDFI.
\item The amplitude of the oscillations in the surface pressure (and mean sea level pressure) was computed from three consecutive coupling files.
\end{itemize}

The last item actually detects situations when the moving pressure disturbance would be missed using $2\Delta t$ (6 hours) coupling update interval not the $\Delta t$ (3 hours) interval. But the large values of this field can mean that the interval as short as $\Delta t$ can also be insufficient for proper representation of lateral boundary data by linear interpolation of the LBC fields in time. 

\subsection{Computing the monitoring of the coupling update frequency (MCUF) field from the ECMWF coupling files}

ALADIN can compute the MCUF field during the model forecast.
The field was computed by running ALADIN on the LACE domain of LBC files from operational IFS with horizontal resolution of 15.4 km (the same resolution and grid as the coupling files) and a time-step of 600 seconds. 
The output IFSM field is written with 3 hourly interval. 
The same procedure has been performed on the LBC files provided since 27th October 2010 until 15th November 2014, for 4 runs per day (starting from 00, 06, 12 and 18 UTC analyses) and extending to 78 hours forecast. 

The maximum value of the IFSM field on the domain covered by the coupling files has been computed for each forecast output file. 
The average IFSM has been computed, the number of files when it exceeded the critical value and the maximum value achieved in each grid point for the coupling files for 6 hours forecast and longer.

\subsubsection{The same procedure applied to the ARPEGE coupling files}

MCUF was also computed by running ALADIN on the domain and resolution of the coupling files from ARPEGE and this fileld is refered to as the ARPM field to distinguish it from the MCUF field computed in ARPEGE forecast. 
But the coupling files from the ARPEGE global model are provided in different horizontal resolutions that the files from IFS. 
There was no period when both coupling files used the same horizontal resolution (Table \ref{table}).
It is more important to test the method on both sets of coupling files on the same period in time since the frequency of the occurence of the fast storms can have significant seasonal and annual variability.

\subsection{The error function}

Each coupling file contains the complete set of model fields that can be also used as a initial file to perform a forecast run using ALADIN model.
The coupling data are used as initial fields to perform a model integration of one time step forward in time in order to obtain $F(t+\delta t)$ and the tendencies of the model variables.
In order to avoid spurious high frequency noise, a filter initialization should be applied before the start of the model run.

When investigating the error due to linear interpolation of surface pressure, \cite{termonia2003} computes an error function from the surface pressure field and finds that its maximum over the model domain is a good indicator of a RMPD.
Each coupling file contains the complete set of fields needed to initialize the model, so they can be used as initial fields to perform one time step model integration.
\cite{termonia2003} defines a dimensionless estimate of the truncation error due to linear interpolation in time as
\begin{equation}
e_T=\frac{1}{4}\left | \frac{\left ( F'(t_2)-F'(t_1)\right ) (t_2-t_1)}{F(t_1)+F(t_2)} \right | .
\label{eqet}
\end{equation}
Where $F(t_{1,2})$ are the values of the model field $F$ at times when the LBC data are available in the coupling files and $t_2-t_1$ is therefore the coupling update interval (3 hours).
$F'(t_{1,2})$ is the tendency of the field $F$ at time $t_{1,2}$ and can be computed as 
$F'(t_{1,2})=\frac{F(t_{1,2}+\delta t)-F(t_{1,2})}{\delta t}$ where $\delta t$ is the model time step.
%$F'(t_1)$ is the tendency of the field $F$ at time $t_1$ and can be computed as 
%$F'(t_1)=\frac{F(t_1+\delta t)-F(t_1)}{\delta t}$ where $\delta t$ is the model time step.
The error function of surface pressure and mean sea level pressure was computed for each coupling file. 
The tendencies can be computed without any filtering of the field in coupling files, using DFI \citep{lynch1997b} or SSDFI \citep{termonia2008}.

The error function $e_T$ has been computed for the surface pressure field from IFS coupling files.
The maximum values over the model domain are
\begin{equation}
E_T=max{ \left ( e_T(x,y) \right ) }
\label{eqEt}
\end{equation}
where $e_T$ is the error computed in each grid point.

The error estimate $E_T$ revealed cases when linear interpolation of the coupling data in time with 3 hour coupling update interval is insufficient for the Belgian domain \citep{termonia2003}.
Both $E_T$ computed with or without filtering over the Belgian domain yield a clear signal when there is a intensive RMPD. But the domain of Aladin Belgium used in that work did not contain any strong orography.
The Croatian domain (and hence the LACE coupling domain) contains mountains of considerable height (Alps, Apennines etc.).

\subsubsection{Digital filter initialization}

Coupling files contain already interpolated data (to a lambert conformal grid), not the data from the native global model grid.
Horizontal interpolation of the surface pressure field (and other forecast fields) from native IFS grid and topography to the grid and topography of the LBC files also distorts the fields, so there could be spin-up when computing the tendencies.
This change in geometry can generate high frequency noise that can be removed using DFI \citep{lynch1992}. 
The DFI was applied using Dolph-Chebyshev filter on 14 time steps adiabatic backward integration and 14 time steps forward integration with a time step of 600 seconds. 
The time span was 2.333 hours, the stop band edge period was 3 hours, the ripple ratio 0.05 yields minimum time span of 2.07 hours \citep{lynch1997a} used with the scheme for diabatic DFI in ALADIN \citep{lynch1997b}.

\subsubsection{Scale selective digital filter initialization}

Doppler effect can shift the frequencies of RMPDs into the range of spurious gravity waves that DFI was designed to remove.
Consequently, DFI reduces the intensity of RMPDs \citep{termonia2008}. 
Alternative SSDFI is expected to be a better solution for initialize the fields used to compute the error function intended to detect RMPDs.

The SSDFI was applied using Dolph-Chebyshev filter on 8 time steps adiabatic backward integration and 8 time steps forward integration with a time step of 600 seconds. 
The time span was 1.333 hours, the stop band edge period was 1.5 hours, the ripple ratio 0.05 yields minimum time span of 1.019 hours and the cutoff frequency increases with wave number for 30 m/s \citep{termonia2008}. 
This shorter time span and stop band edge period yields less filtering that preserves the storm in \cite{termonia2008} while still removing the spurious inertia gravity waves generated above mountains.
Shorter time span means shorter model run which is also beneficial in the operational context.

Both filtering methods require running the model adiabaticaly backwards for a number of time-steps and then diabatically forward for the same number of time steps for each of the coupling files. The method is therefore computationally expensive if DFI or SSDFI are applied before computing the tendencies (about as expensive as IFSM).

\subsection{The amplitude in the pressure variations}

All the methods described previously require that all the coupling files (initial and forecast) contain the data necessary to initialize the LAM and run the LAM at least for one time step. 
Here a very simple method for detecting RMPDs is presented that does not require running LAM.

As a measure of variability in the model field, the following can be computed:
\begin{equation}
A=\frac{1}{2}\left ( F(t_1)+F(t_3)-2F(t_2) \right )
\label{eqa}
\end{equation}
where $F(t_1)$, $F(t_2)$ and $F(t_3)$ are the values of the model field $F$ at three consecutive times $t_1$, $t_2$ and $t_3$ when the coupling data are available. The differences in times is the coupling update interval $t_2-t_1=t_3-t_2=\Delta t$ which is operationally equal to 3 hours. 

Eq.\ref{eqa} describes the changes of the model field $F$ during the $2\Delta t$ period, eg. twice the coupling update period.
Therefore, the values of $A$ are largest in points where $\Delta t$ period is actually enough to describe the evolution of the model variable during the coupling update interval using linear interpolation in time (eg. at the position of the pressure minimum at time $t_2$).
However, A can be used as an indicator of a RMPD, as will be shown in the results of this study.
On the other hand, $A$ could miss the evolution of the model variable on a time scale less than $\Delta t$, for example when the model variable evolves as the full line in Fig 1 of Termonia (2003).

\subsection{The effect of linear interpolation }

An atmospheric disturbance can enter the domain unnoticed by the coupling scheme. The Figure \ref{mslp1} shows mean sea level pressure from the ARPEGE forecast (as provided in the coupling file) and mean sea level pressure from the ALADIN 8km forecast coupled to it.

Linear interpolation in time distorts the model fields. 
Figure \ref{intmslp} shows the effect of linear interpolation on the mean sea level pressure. 
The ARPEGE forecast mean sea level pressure from two consecutive coupling files is interpolated linearly in time (as in the operational coupling procedure). 
In the place of moving storm, LAM sees a dual cyclone structure, one cyclone/storm disappears and another appears. This is why larger coupling zone yields dual cyclone structure, as was shown by \citet{tudor2010b}.

Other meteorological fields that are used for coupling at lateral boundaries get distorted by linear interpolation in time if they contain high resolution features such as storms or meteorological fronts. For simplicity, this article will focus on the mean sea level pressure and surface pressure fields.

\section{Filtered surface pressure field from ARPEGE}

\subsection{The time series of MCUF maxima}

The maximum value of the MCUF field as computed in the operational ARPEGE has been extracted from each forecast coupling file available for the whole LACE coupling domain. 
The time series of MCUF maxima are shown in Figure \ref{mcufmaxar}.
The MCUF maxima from the 3 hour forecast files were omitted in the plot since they had high values due to other phenomena that arose during spin-up following ARPEGE initialization, especially in the period until 6th February 2008.
Most of the points with large MCUF values in the 3 hour ARPEGE forecast are close to mountains.
This suggests large spin-up of the surface pressure field in the beginning of the ARPEGE forecast.
Since these large values of MCUF in the +03 hour forecast mostly do not represent a storm
that moves quickly through the domain, analysis has been performed only on fields from +06 hour forecast or larger.

MCUF exceeds the 0.003 value rather often, mostly in events that last a few days, up to a week. 
For each file where MCUF was larger than this threshold value, a figure was plotted with mean sea level pressure from the coupling file (ARPEGE) and the operational ALADIN forecast in 8 km resolution coupled to it, and the points where MCUF was larger than 0.003 (see example in Figure \ref{mslp1}).
Each time, large MCUF values were associated to a pressure disturbance in ARPEGE that was often less intensive in ALADIN forecast (if covered by the operational ALADIN domain).

The events that yield large values of the MCUF field represent RMPDs that rapidly traverse any part of the LACE domain.
These events are more frequent in autumn, but appear throughout the year, least often during summer months.
Several large MCUF values can be associated to a single event (one cyclone moving rapidly over the model domain), but they represent maxima from different forecast coupling files and different forecast runs (starting from different initial times corresponding to different ARPEGE analyses).
On the whole LACE domain, the critical value of 0.003 has been exceeded 3045 times in 288648 files, more than 1\% of the files in the whole period from 23rd January 2006 until 16th November 2014 (see Table \ref{table}). In 878 files, large MCUF values were close to the coupling zone of the operational ALADIN domain in DHMZ (see Figure \ref{mslp1}). 
This is only 0.3\% of the coupling files and the event can be considered rare. But, as mentioned earlier, these events are perhaps most important to be forecast.
In order to properly forecast such events using LAM, one should first detect it and then apply boundary error restarts \citep{termonia2009} or gridpoint nudging \citep{termonia2011}.

\subsection{Spatial distribution of MCUF from ARPEGE}

Successfull implementation of the computations of the MCUF field in the operational ARPEGE means that it is not dependent on the horizontal resolution of the global model since ARPEGE is run on a stretched grid.
The averaged MCUF fields (Figure \ref{mcufaravg}) for different horizontal resolutions (Figure \ref{mcufaravg}a for 20.678km, Figure \ref{mcufaravg}b for 15.4 km and Figure \ref{mcufaravg}c for 10.51 km) show that it does not depend on the resolution of the coupling files as well as the resolution of the global model where it was computed. Averaged MCUF field is slightly larger over the North Sea in the first period (from 23rd January 2006 until 6th February 2008) for the lowest resolution. The values over the Mediterranean have the highest values in the middle period (from 6th February 2008 until 11th May 2010) for the 15.4 km resolution of the coupling files. 
This result suggests that the cyclones traversed Mediterranean more often and faster during that period than in the periods before and after.
%Alternatively, it is also possible that the set-up of the operational ARPEGE allowed the development of deeper and faster cyclones over the Mediterranean (but not simultaneously over the North Sea and the Black Sea).

The maps of number of cases when the MCUF field exceeded the 0.003 threshold (Figure \ref{mcufarnum}) show that the number of cases with fast cyclones over the North Sea is the largest in the last period (that is also twice as long as the other two). But over the Mediterranean, MCUF exceeded the critical value most often in the second period, as well as over the area under the influence of the Bay of Biscay.

The absolute maximum values of the MCUF field have large values over most of the western Mediterranean during the second period (Figure \ref{mcufarabs}). 
The overall largest values of MCUF were computed during the third period (and in the highest spatial resolution) close to the coastline of Algeria, but the values are low over the rest of the Mediterranean. 
On the other hand, the maxima are the highest over the North sea in the last period and over the Black Sea in the first period. 
%These results suggest shifting of the area with the largest cyclone intensity and speed from one side to the other.

The spatial distribution of the frequency of the events when MCUF exceeded the critical value (Figure \ref{mcufarnum}) indicate which areas should be avoided as parts of the coupling zone
if one wants to have fewer problems with properly resolving the boundary data in time with 3 hourly coupling update period.
When the filtered surface pressure field is larger than a threshold value 0.003, there is a storm rapidly propagating through the area. If the point with the large  value is inside the coupling zone of a LAM, it can be expected that the LAM forecast will miss the storm due to time interpolation of boundary data. 
The analysis of the MCUF field from ARPEGE coupling files for the common LACE coupling domain shows that this field is above the threshold far more frequently than acceptable. 

\section{Detecting rapidly moving pressure disturbances (RMPDs) in the ECMWF coupling files}

MCUF is not computed by operational IFS, the alternative methods of detecting RMPDs have been tested on the coupling files received operationally from ECMWF.

\subsection{Computing MCUF by running ALADIN model on the coupling files from IFS}

MCUF computed by running ALADIN in the resolution of the coupling files from IFS using interpolated IFS analysis as the initial conditions (without any filtering) for 4 runs per day up to 78 hours forecast with 3 hourly output.
The MCUF field computed this way is referred to as IFSM.
The initial IFSM values are zero. IFSM computed during the first 3 hours of forecast has very large values due to model spin-up so only the fields corresponding to the 6 hour forecast and longer are used in the analysis.

\subsubsection{The time series of IFSM maxima}

The time series of the maximum values of IFSM field from the whole LACE domain for forecast ranges from 6 to 78 hours are shown in Figure \ref{mcufmaxec} for the period from 27th October 2010 until 15th November 2014. 
The critical value is exceeded in 698 files (out of total 147350 files) during the 4 year period and over the whole domain (see Table \ref{table}). 
This is less often than in ARPEGE, since during the same period MCUF was larger than 0.003 in 995 files (out of 129674 files).
The total number of files is larger for IFS than for ARPEGE since ARPEGE forecast LBC files extend up to 72 hours (and only 60 hours for the 18 UTC run), while files from all runs of IFS extend up to 78 hour forecast. 

Although the critical value of 0.003 is exceeded less often with IFSM than with MCUF in ARPEGE, there are periods with large values associated to RMPDs during every part of the year, more often in autumn and the least often in summer.
A figure with mean sea level pressure from the IFS coupling file and gridpoints with large IFSM values were plotted for each coupling file for which IFSM exceeded the critical value in order to estimate if the large IFSM values are associated to the cyclones in the IFS files (and not only in the ALADIN forecast run used to compute the IFSM field).
Inspection of this set of figures lead to a conclusion that large values of IFSM are connected to a pressure low in IFS fields.

One should keep in mind that the MCUF values are computed by running ALADIN using IFS coupling files (initial and forecast). 
ALADIN model can yield different evolution of model variables, including surface pressure, so that large MCUF values correspond to a cyclone that moves quickly in the ALADIN forecast, not neccessarily in the IFS forecast. 
On the other hand, a RMPD in the IFS forecast might be less intensive or slower in the ALADIN forecast due to differences in the model set-up, choices in physics and dynamics.

\subsubsection{Spatial distribution of IFSM}

MCUF was computed by running ALADIN forecast on a limited area domain in 15.4 km resolution. Coupling zone was 8 points wide. The procedure could have missed a cyclone entering the LACE domain during the coupling interval. 
It is also expected to get unwanted phenomena in the IFSM field in the coupling zone of LBC files. 

In the figure \ref{mcufecavg}, a small dot is plotted in the position of each model grid-point in the colour corresponding to the average IFSM value multiplied by 1000 as shown in the colour scale below. 
Average IFSM field and average MCUF from ARPEGE for the same period (Figure \ref{mcufecavg}) have substantially different spatial distributions. 
The differences are most pronounced over the Baltic area, where IFS yields more fast cyclones and over Mediterranean, where ARPEGE forecasts more RMPDs. 

Maximum MCUF has larger values than IFSM (Figure \ref{mcufecabs}). 
The average values are low along lateral boundaries, but the maxima do not decrease towards the lateral boundaries (Figure \ref{mcufecavg}).
The differences in the maximum MCUF and IFSM values are much less pronounced than for the averaged fields. 

In most of the domain, MCUF and IFSM exceeded the critical value less than once in the 4 year period (Figure \ref{mcufecmap}). 
The most critical part is in the north, where cyclones apparently traverse rather quickly and the number of files where IFSM s larger than threshold threshold exceeds 20. 
Both MCUF and IFSM show areas where pressure disturbances move more rapidly and/or frequently than elsewhere, such as the North Sea, the Baltic, western Mediterranean and west coast of the Black Sea.
The critical value of 0.003 is exceeded more often for IFSM than in ARPEGE (Figure \ref{mcufecmap}), over the North Sea, western Black Sea and the Baltic, but less often over the western Mediterranean.
This suggests that IFSM field could be missing some of the RMPDs approaching Adriatic Sea and Croatia over the western Mediterranean.

\subsubsection{Computing MCUF by running ALADIN model on the coupling files from ARPEGE}

ARPM was computed by running ALADIN on the domain and resolution (10.61km) of the ARPEGE coupling files with 450 seconds time step starting from the ARPEGE analysis without initialization.
The time series of ARPM maxima over the LBC domain are shown in Figure \ref{mcufecar}.
There is a good agreement with MCUF computed in ARPEGE.
But ARPM gives additional strong signal for the storm that hit Turkey on 27th September 2014. 
MCUF did not show a signal for the same case.

\subsection{The error function values using mean sea level pressure from ECMWF coupling files}

ALADIN was run for one time step using fields from the coupling files from IFS as initial conditions in order to estimate the tendency of the model variables (in particular the surface pressure).
The run is performed on the grid of the coupling files using 600 second time step.
The error is estimated according to equation \ref{eqet} and its maximum over the model domain according to the equation \ref{eqEt}.
The error function was computed for the period since 27th October 2010 until 15th November 2014 for experiments without initialization and initialized with SSDFI, and for the period since 1st January 2013 for the experiment with DFI.

\subsubsection{Tendencies computed without filtering initialization}

%Results show a number of episodes with high values of $E_T$.
The time series of $E_T$ computed without initialization is plotted in Figure \ref{errec}.
The noise is more intensive than with IFSM, but the signal of RMPDs can be seen. 
The level of noise is lower in summer than in winter and it is lower when the error function is computed using mean sea level pressure than for surface pressure.
Due to rather high level of noise, a critical value larger than 0.003 should be defined in order to avoid false alarms.
The method using error estimate sometimes yields large values over mountainous areas. 
If the model domain is defined so that the mountains are not in the intermediate zone (close to lateral boundaries), these events could be ignored by the operational procedure and would not be false alarms.

\subsubsection{Tendencies computed with digital filter initialization}

The time series of $E_T$ computed for fields initialized with DFI is plotted in Figure \ref{serrec} for the period from 1st January 2013 until December 2014.
The noise is much lower than for the test without initialization, but the signal of RMPDs is also weaker.
There is more noise in $E_T$ computed for mean sea level pressure than for surface pressure in winter and spring, but less in the autumn. 
The signal of the RMPDs is removed almost completely from the error function computed for surface pressure, especially in winter and spring.
%DFI reduces the signal of RMPDs.

There is a signal for RMPD in $E_T$ computed from mean sea level pressure on 27th November 2013 that does not exist in the time series of $E_T$ for the surface pressure. The peak is located over the Alps and shows preristently for model runs from successive analyses about the same time (9 to 15 UTC that day). The satellite figures of the area for that date show clouds associated to mountain waves (not shown).

\subsubsection{Tendencies computed with scale selective digital filter initialization}

Similarly, the error function was computed after the fields in the coupling files have been initialized using SSDFI for the period since 27th October 2010 until December 2014.
The time series of the maxima of the error function is ploted in Figure \ref{sserrec}.
The level of noise and the intensity of the signal of approaching RMPDs are similar to tose computed with DFI. But there are subtle differences. Several cases of RMPDs are more pronounced and there is no signal on 27th November 2013 that occured when DFI was used.

\subsection{Amplitude of oscillations in mean sea level pressure}

The amplitude of oscillations in mean sea level pressure was computed for the coupling files from IFS for the period since 27th October 2010 and for the coupling files from ARPEGE since 1st January 2013, both until December 2014.
The time series of the maxima in the amplitude of the mean sea level pressure variations from IFS is displayed in Figure \ref{ampec} and for ARPEGE in Figure \ref{ampar}.

Although the amplitude maxima achieve large values during periods without RMPDs (the periods without RMPDs are those when MCUF and IFSM are low), the amplitude is so much larger in a case with RMPD that there is a signal that can be distinguished in the noisy pattern.

A figure was plotted with mean sea level pressure from the coupling file from IFS and all points with large values of $A$ ($A>0.003$) for each case when this threshold was exceeded.
The majority of the cases are related to propagating cyclones and pressure throughs and are usually associated to the large values of IFSM.
However, there are cases when $A$ is larger than the threshold in mountainous regions of Alps, Atlas mountains and Turkey, but these are associated to an atmospheric front approaching the area so the large values could not be dismissed as false.

There is also a number of cases when IFSM did not indicate a RMPD, while $A$ did reach values above the threshold in points close to the edge of the coupling domain. 
The subsequent coupling times also had large values of $A$ in the vicinity.
In these cases, the cyclone entered the coupling domain too quickly to be detected by the procedure used to compute the IFSM field.


\conclusions  %% \conclusions[modified heading if necessary]


The three hourly coupling update interval is insufficient for resolving the storm in lateral boundaries as presented for the Lothar storm case \citep{termonia2003}. 
\cite{davies2014} recommends choosing carefully the resolution and fequency of large scale LBCs. However, meteorological services that depend on LBCs from elsewhere might have little choice.
A coupling update frequency is sufficient if the large scale model data contains only features that are large enough and slow enough to be resolved by the coupling update period \citep{denis2003}.
Therefore, the coupling update frequency is determined by the properties of the global model, not the LAM that uses it for LBCs.

\cite{termonia2004} developed a strategy to monitor rapid changes in surface pressure in ARPEGE by producing a diagnostic output field for the filtered surface pressure (MCUF).
This field is provided in the coupling files since 06 UTC run on 23rd January 2006 for the LACE coupling domain.

When MCUF is larger than a threshold value of 0.003 \citep{termonia2004}, there is a rapid development in the surface pressure suggesting that a fast cyclone has moved through the area. If the point with the large  value is inside the coupling zone of the ALADIN domain, it can be expected that the ALADIN model run will miss the cyclone strength and development due to  time interpolation of boundary data.
When the time series of MCUF data has been analysed for the Belgian domain \citep{termonia2009}, it was found that such events occurred only several times per year.

The analysis of the MCUF field in this article shows that this field is above the threshold more frequently for the whole LACE coupling domain as well as for the coupling zone of the Croatian operational domain (it covers larger area than the operational Belgian domain in \citep{termonia2003}), but the event can still be considered rare. 
There are changes from one season to another (more ot less 'stormy'), but there is no apparent increase in the number of fast propagating storms with an increase of the ARPEGE resolution (at least in the range of resolutions available for this study).

The spatial distribution of MCUF reveals that RMPDs favour the sea surfaces, especially the North Sea and the western Mediterranean. 
Analysis of the MCUF and IFSM fields for a longer period can show which areas favour quickly moving storms that could be missed by the coupling procedure if the 3 hourly coupling period is used. 
Maps with number of occurrences when the filtered pressure field is larger than the 0.003 threshold show that there are not to many places where to put the coupling zone in order to avoid LAM forecast failure in the case of a RMPD. 
The problem would be only made worse in higher resolution LAM. The coupling zone on the lateral boundaries is 8 grid points wide and shrinks with the resolution increase. The storm needs less time to cross the narrow coupling zone.
Higher resolution global model can yield more intensive pressure changes.

The spatial distribution can be viewed as a map of the fast cyclone tracks and areas that support rapid changes in cyclone development.
Not surprisingly, this study shows that not only North Sea, but also the western Mediterranean is an area where storms frequently propagate with high velocities and can not be resolved in LBCs of a 8 km resolution LAM when provided with 3 hour interval.
In LAM with roughly 3 times larger horizontal resolution, even 1 hour coupling interval would be insufficient.

There is no field similar to MCUF provided in the coupling files of IFS from ECMWF. 
Therefore an experiment has been performed in order to compute the field locally from the coupling files. 
The forecast needed to compute MCUF was run using ALADIN model and the resulting field IFSM can be used for detecting RMPDs in the operational forecast. It requires running the ALADIN forecast in low resolution up to 78 hours (same range as the coupling files are provided). It is more computationally  expensive than reading the field already provided in the LBC files but it is feasible. 
However, the results contain some detrimental effects:
\begin{itemize}
\item different model dynamics could lead to different developments in the surface pressure field and hence different MCUF values,
\item a quickly moving storm can enter the LBC domain undetected and consequently be missed by the MCUF,
\item rather low cyclone activity on the western Mediterranean comapred with results using ARPEGE.
\end{itemize}
%There are at least two possible reasons for this. The cyclogenesis in the lee of the mountains of the northern Africa is low in IFS (lower than in ARPEGE). Or (and more likely) running ALADIN on the coupling file domain can not model the cyclones forming in the lee of north African mountains since much of these mountains are outside of the domain.

The error function \citep{termonia2003} were computed using tendencies estimated by running ALADIN for one time step, using fields from the coupling fields without initialization, initialized with DFI and with SSDFI.
No initialization yields a signal of RMPDs but also a lot of noise. Clearly a higher threshold value should be used, but it should be chosen carefully.
DFI reduces the level of noise and the magnitude of the signal and many RMPDs are removed from the time series (Figure \ref{serrec}) but there are still evidences of large values related to mountains.
SSDFI reduces the level of noise and the signal of RMPDs, but more of the signal is preserved.

Finally, RMPDs are detected by simple computations of variations in the mean sea level pressure from three consecutive coupling files.
Apparently, this rather simple method can be used for detecting RMPDs. The noise is more intensive than for error function computed without initialization, but so is the signal for RMPDs. 
This method can be used on any variable and it does not require running any model using coupling data as initial conditions.
Mean sea level pressure is less sensitive to the reduction in the coupling update frequency than precipitation and vorticity \citep{denis2003}.

Climate LAMs could benefit from a large domain \citep{zagar}.
It takes several days for the cascade of variance to fill the small scales \citep{laprise2008}. 
Loosing small scale features, arriving from the global model at lateral boundaries, certainly does not help.
If the domain of the climate LAM is small and the flow over the area is strong, it could move over the domain too quickly to develop small scales \citep{zagar}, and if the temporal interpolation of LBC data filters high resolution data from a global model, there might not be enough space (in the domain) nor time (before the flow leaves it) for LAM to recreate these small scales.

On the other hand, NWP models that have small scale data in the initial conditions through blending \citep{brozkova} or data assimilation cycle (eg. \cite{stanesic2011}) need RMPDs that enter the domain during the model forecast. It took ALADIN 66 hours to develop a small scale feature in the 2km resolution nonhydrostatic run \citep{tudor2010} coupled to 8km operational forecast that was run without data assimilation at the time \citep{ivateks2004}.

As there are plans to increase the resolution of the operational ALADIN to 4km and ECMWF announced plans for the increase in the horizontal resolution of operational IFS, the problem of resolving RMPDs in LBC data available with 3 hourly interval will become more frequent and it is questionable if hourly coupling data would be sufficient in some cases.
Boundary error restarts \citep{termonia2009}, gridpoint nudging \citep{termonia2011}, computing corrected interpolation in time with time derivatives \citep{termonia2003} and alternative methods of interpolating LBC data in time \citep{tudor2010b} are computationally expensive and should be used only when needed.
Therefore such cases should be detected by a reliable method since any missed case means that LAM would not forecast severe weather conditions.
The error function computed without initialization and the amplitude method (Section 4.3) are cheap methods that could be applied in a straightforward manner. MCUF from IFSM seems reliable for most of the LACE domain. The error function computed from the initialized fields does not improve the results enough to justify the extra computational cost.
The alternative is to compute MCUF in operational IFS.


%\appendix
%\section{}    %% Appendix A

%\subsection{}                               %% Appendix A1, A2, etc.




\begin{acknowledgements}
The author is gratefull to the two anonymous reviewers whose constructive comments have improved the manuscript.
\end{acknowledgements}


%% REFERENCES

%% The reference list is compiled as follows:



%% Since the Copernicus LaTeX package includes the BibTeX style file copernicus.bst,
%% authors experienced with BibTeX only have to include the following two lines:
%%
%% \bibliographystyle{copernicus}
%% \bibliography{example.bib}
%%
%% URLs and DOIs can be entered in your BibTeX file as:
%%
%% URL = {http://www.xyz.org/~jones/idx_g.htm}
%% DOI = {10.5194/xyz}


%% LITERATURE CITATIONS
%%
%% command                        & example result
%% \citet{jones90}|               & Jones et al. (1990)
%% \citep{jones90}|               & (Jones et al., 1990)
%% \citep{jones90,jones93}|       & (Jones et al., 1990, 1993)
%% \citep[p.~32]{jones90}|        & (Jones et al., 1990, p.~32)
%% \citep[e.g.,][]{jones90}|      & (e.g., Jones et al., 1990)
%% \citep[e.g.,][p.~32]{jones90}| & (e.g., Jones et al., 1990, p.~32)
%% \citeauthor{jones90}|          & Jones et al.
%% \citeyear{jones90}|            & 1990



%% FIGURES

%% ONE-COLUMN FIGURES

%%f
%\begin{figure}[t]
%\includegraphics[width=8.3cm]{FILE NAME}
%\caption{TEXT}
%\end{figure}
%
%%% TWO-COLUMN FIGURES
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../FP/mslp_coupl200810271260.eps}
\caption{Mean sea level pressure  (hPa)from ARPEGE (green) and ALADIN (red) operational 60 hour forecast starting from 12 UTC analysis on 27$^{th}$ Oct 2008. The coordinates and values of MCUF field exceeding the 0.003 threshold are listed in the upper right corner and plotted as blue dots on the map.}
\label{mslp1}
\end{figure*}
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../FP/mslp2008103000int2.eps}
\caption{Mean sea level pressure (hPa) from ARPEGE operational coupling files starting from 12 UTC analysis on 27$^{th}$ Oct 2008, 57 (a) and 60 (i) hour forecasts, linear interpolation of mean sea level pressure in time to half of the 3 hour coupling period (e), 1/8 of 3h (b), 1/4 (c) 3/8 (d), 5/8 (f), 3/4 (g) and 7/8 (h).}
\label{intmslp}
\end{figure*}
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/mcufmax.eps}
\caption{Maximum value of the MCUF field (units hPa) on the LACE coupling domain, provided from ARPEGE, from the coupling files for 6 hour forecast up to 72 hours forecast (60 hours for 18 UTC run), starting from 00, 06, 12 and 18 UTC analyses, since 23rd January 2006 until 15th November 2014.}
\label{mcufmaxar}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapavg1.eps}
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapavg2.eps}
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapavg3.eps}
\caption{Average MCUF field (units 0.001 hPa) from ARPEGE for different resolutions of the LACE coupling files: 
(a) 20.678 km averaged for the period 23$^{rd}$ Jan 2006 to 6${th}$ Feb 2008.
(b) 15.4 km averaged for the period 6$^{th}$ Feb 2008 to 11$^{th}$ May 2010.
(c) 10.51 km averaged for the period 11$^{th}$ May 2010 to 15$^{th}$ Nov 2014.}
\label{mcufaravg}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapnum1.eps}
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapnum2.eps}
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapnum3.eps}
\caption{The number of times the MCUF field from ARPEGE exceeds 0.003 threshold for different resolutions of the coupling files: 
(a) 20.678 km averaged for the period 23$^{rd}$ Jan 2006 to 6${th}$ Feb 2008.
(b) 15.4 km averaged for the period 6$^{th}$ Feb 2008 to 11$^{th}$ May 2010.
(c) 10.51 km averaged for the period 11$^{th}$ May 2010 to 15$^{th}$ Nov 2014.}
\label{mcufarnum}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapabs1.eps}
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapabs2.eps}
\includegraphics[angle=270,width=8.3cm]{../grads/mcufmapabs3.eps}
\caption{Absolute maximum values of the MCUF field (units 0.01 hPa) from ARPEGE for different resolutions of the coupling files:
(a) 20.678 km averaged for the period 23$^{rd}$ Jan 2006 to 6${th}$ Feb 2008.
(b) 15.4 km averaged for the period 6$^{th}$ Feb 2008 to 11$^{th}$ May 2010.
(c) 10.51 km averaged for the period 11$^{th}$ May 2010 to 15$^{th}$ Nov 2014.}
\label{mcufarabs}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/mcufecmax.eps}
\caption{Time series of maximum value of IFSM field (units hPa) on the coupling LACE domain for 6 hour forecast up to 78 hours forecast, computed by running ALADIN, starting from 00, 06, 12 and 18 UTC analyses, since 1st November 2010 until 15th November 2014.}
\label{mcufmaxec}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../grads/mcufmapavgec.eps}
\includegraphics[angle=270,width=12cm]{../grads/mcufmapavgar.eps}
\caption{Spatial distribution of the average IFSM (top) and MCUF (bottom) values (units 0.001 hPa) for forecast hour greater than or equal to 06 hours for the period since 1$^{st}$ Nov 2010 until 15$^{th}$ Nov 2014. }
\label{mcufecavg}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../grads/mcufmapabsec.eps}
\includegraphics[angle=270,width=12cm]{../grads/mcufmapabsar.eps}
\caption{Spatial distribution of the maximum of absolute IFSM (top) and MCUF (bottom) (units 0.01 hPa), for forecast hour greater than or equal to 06 hours for the period since 1$^{st}$ Nov 2010 until 15$^{th}$ Nov 2014. }
\label{mcufecabs}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../grads/mcufmapnumec.eps}
\includegraphics[angle=270,width=12cm]{../grads/mcufmapnumar.eps}
\caption{Spatial distribution of the number of occurences when IFSM (top) and MCUF (bottom) values exceed the value 0.003, for forecast hour greater than or equal to 06 hours for the period since 1$^{st}$ Nov 2010 until 15$^{th}$ Nov 2014. }
\label{mcufecmap}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/mcufecar.eps}
\caption{Time series of maximum value of ARPM (MCUF computed by running ALADIN on the coupling LACE domain from ARPEGE (the domain and resolution of LBC files) with 450 sec time-step).}
\label{mcufecar}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/errecmslp.eps}
\caption{Time series of maximum value of error function ($E_T$, Eq. \ref{eqEt}) without any filtering initialization.}
\label{errec}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/serrecmslp.eps}
\caption{Time series of maximum value of error function, fields are initialized with DFI.}
\label{serrec}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/sserrecmslp.eps}
\caption{Time series of maximum value of error function, fields are initialized with SSDFI.}
\label{sserrec}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/mcufampec.eps}
\caption{Time series of the maximum value of the amplitude in the mean sea level pressure variations (Eq. \ref{eqa}) computed from the coupling files from IFS.}
\label{ampec}
\end{figure*}
%
%f
\begin{figure*}[t]
\includegraphics[angle=270,width=12cm]{../gnuplot/mcufampar.eps}
\caption{Time series of the maximum value of the amplitude in the mean sea level pressure variations (Eq. \ref{eqa}) computed from the coupling files from ARPEGE.}
\label{ampar}
\end{figure*}
%
%%f
%\begin{figure*}[t]
%\includegraphics[width=12cm]{FILE NAME}
%\caption{TEXT}
%\end{figure*}
%
%
%%% TABLES
%%%
%%% The different columns must be seperated with a & command and should
%%% end with \\ to identify the column brake.
%
%%% ONE-COLUMN TABLE
%
%%t
%\begin{table}[t]
%\caption{TEXT}
%\begin{tabular}{column = lcr}
%\tophline
%
%\middlehline
%
%\bottomhline
%\end{tabular}
%\belowtable{} % Table Footnotes
%\end{table}
%
%%% TWO-COLUMN TABLE
%
\begin{table*}[t]
\caption{Model, period, horizontal resolution and total number of the coupling files for which the rapid changes of surface pressure field were analyzed, the field was used received from Meteo-France and computed by ALADIN for files received from ECMWF. The rapid changes in surface pressure for the first 3 hours were ommited from the analysis due to evidence of model spin-up for some periods.}
\vskip4mm
\centering
\begin{tabular}{| c c c c c c c c |}
\tophline
model & period  & resolution & total num  &  whole & domain & MCUF & MCUF$>0.003$ \\
 & (from-to) & (km) & of files & $>0.003$ & $>0.004$ & $>0.005$ & coupling zone \\
\middlehline
ARPEGE & 06Z23Jan2006 -- 00Z06Feb2008 & 20.678 & 64292 & 906 & 270 & 93 & 235 \\
\middlehline
ARPEGE & 06Z06Feb2008 -- 00Z11May2010 & 15.400 & 72600 & 1017 & 383 & 141 & 400 \\
\middlehline
ARPEGE & 06Z11May2010 -- 00Z16Nov2014 & 10.610 & 151756 & 1122 & 293 & 125 & 243 \\
\middlehline
ARPEGE & 06Z23Jan2006 -- 00Z16Nov2014 & all & 288648 & 3045 & 946 & 359 & 878 \\
\middlehline
ARPEGE & 06Z01Nov2010 -- 00Z16Nov2014 & 10.610 & 129674 & 995 & 259 & 108 & 186 \\
\middlehline
IFS & 06Z01Nov2010 -- 00Z16Nov2014 & 15.400 & 147350 & 698 & 178 & 67 & 109 \\
\bottomhline
\end{tabular}
\label{table}
\end{table*}
%%t
%\begin{table*}[t]
%\caption{TEXT}
%\begin{tabular}{column = lcr}
%\tophline
%
%\middlehline
%
%\bottomhline
%\end{tabular}
%\belowtable{} % Table Footnotes
%\end{table*}
%
%
%%% NUMBERING OF FIGURES AND TABLES
%%%
%%% If figures and tables must be numbered 1a, 1b, etc. the following command
%%% should be inserted before the begin{} command.
%
%\addtocounter{figure}{-1}\renewcommand{\thefigure}{\arabic{figure}a}
%
%
%%% MATHEMATICAL EXPRESSIONS
%
%%% All papers typeset by Copernicus Publications follow the math typesetting regulations
%%% given by the IUPAC Green Book (IUPAC: Quantities, Units and Symbols in Physical Chemistry,
%%% 2nd Edn., Blackwell Science, available at: http://old.iupac.org/publications/books/gbook/green_book_2ed.pdf, 1993).
%%%
%%% Physical quantities/variables are typeset in italic font (t for time, T for Temperature)
%%% Indices which are not defined are typeset in italic font (x, y, z, a, b, c)
%%% Items/objects which are defined are typeset in roman font (Car A, Car B)
%%% Descriptions/specifications which are defined by itself are typeset in roman font (abs, rel, ref, tot, net, ice)
%%% Abbreviations from 2 letters are typeset in roman font (RH, LAI)
%%% Vectors are identified in bold italic font using \vec{x}
%%% Matrices are identified in bold roman font
%%% Multiplication signs are typeset using the LaTeX commands \times (for vector products, grids, and exponential notations) or \cdot
%%% The character * should not be applied as mutliplication sign
%
%
%%% EQUATIONS
%
%%% Single-row equation
%
%\begin{equation}
%
%\end{equation}
%
%%% Multiline equation
%
%\begin{align}
%& 3 + 5 = 8\\
%& 3 + 5 = 8\\
%& 3 + 5 = 8
%\end{align}
%
%
%%% MATRICES
%
%\begin{matrix}
%x & y & z\\
%x & y & z\\
%x & y & z\\
%\end{matrix}
%
%
%%% ALGORITHM
%
%\begin{algorithm}
%\caption{�}
%\label{a1}
%\begin{algorithmic}
%�
%\end{algorithmic}
%\end{algorithm}
%
%
%%% CHEMICAL FORMULAS AND REACTIONS
%
%%% For formulas embedded in the text, please use \chem{}
%
%%% The reaction environment creates labels including the letter R, i.e. (R1), (R2), etc.
%
%\begin{reaction}
%%% \rightarrow should be used for normal (one-way) chemical reactions
%%% \rightleftharpoons should be used for equilibria
%%% \leftrightarrow should be used for resonance structures
%\end{reaction}
%
%
%%% PHYSICAL UNITS
%%%
%%% Please use \unit{} and apply the exponential notation


\end{document}
